Author's response to reviews

Title: Evaluation of PCR procedures for detecting and quantifying Leishmania donovani DNA in large numbers of dried human blood samples from a visceral leishmaniasis focus in Northern Ethiopia.

Authors:

Ibrahim Abbasi (ibrahima@ekmd.huji.ac.il)
Samar Aramin (samar.aramin@gmail.com)
Asrat Hailu (hailu_a2004@yahoo.com)
Welelta Shiferaw (weleltas@gmail.com)
Ayshashm Kassahun (ayshashek2000@yahoo.com)
Shewaye Belay (shewayebelay@yahoo.com)
Charles Jaffe (cjaffe@cc.huji.ac.il)
Alon Warburg (alonw@ekmd.huji.ac.il)

Version: 5 Date: 6 March 2013

Author's response to reviews:

1. The details of the preventive and quality control measures provided in the response to Comment #3.
2. The information on the ITS1 sequences provided in the response to Comment #5.

Both suggestions accepted and added. Visible in track-changes mode